Author's response to reviews

Title: Factors Associated with Default from Treatment among Tuberculosis Patients in Nairobi Province, Kenya: A Case Control Study

Authors:

Bernard N Muture (bnmuture@yahoo.com)
Margaret N Keraka (mnyanchoka2000@yahoo.com)
Peter K Kimuu (pkimu@yahoo.com)
Ephantus W Kabiru (ewkabiru@yahoo.com)
Victor Ombeka (ombeka@yahoo.com)
Francis Oguya (foguya@yahoo.com.au)

Version: 8 Date: 23 August 2011

Author's response to reviews:

The manuscript has been revised and checked for errors and close attention paid to language. Merged cells in the tables have been removed and figure titles listed after the references. Discussion section has been edited and revised. All authors have read and approved the final manuscript.